Author's response to reviews

Title: A Survey of Health Professions Students for Knowledge, Attitudes, and Confidence about Tuberculosis, 2005

Authors:

Marguerite Jackson (majackson@ucsd.edu)
Shawn Harrity (sharrity@ucsd.edu)
Helene Hoffman (hhoffman@ucsd.edu)
Antonino Catanzaro (acatanzaro@ucsd.edu)

Version: 4 Date: 18 July 2007

Author's response to reviews:

Revisions have been made and manuscript has been accepted pending request for additional information re: contributions by authors. The contributions by authors section of the manuscript has been revised as requested.

Marguerite Jackson, PhD